Author's response to reviews

Title: In Vitro Anti-proliferative Activity of Argemone gracilenta and Identification of Some Active Components

Authors:

Juan C Galvez-Ruiz (anisolborano@guayacan.uson.mx)
Mario A Leyva-Peralta (mario_ups@hotmail.com)
Ramon E Robles-Zepeda (rrobes@guayacan.uson.mx)
Adriana Garibay Escobar (agaribay@guayacan.uson.mx)
Laura P Alvarez-Berber (lalvarez@uaem.mx)
eduardo Ruiz-Bustos (erbustos@guayacan.uson.mx)

Version: 2  Date: 12 June 2014

Author's response to reviews: see over
Dear Section Editor,

We are submitting a improved version of the manuscript entitled “In Vitro Anti-proliferative Activity of Argemone gracilenta and Identification of Some Active Components” written by Mario Alberto Leyva-Peralta, Ramón Enrique Robles-Zepeda, Adriana Garibay-Escobar, Eduardo Ruiz-Bustos, Laura Patricia Alvarez-Berber and Juan Carlos Gálvez-Ruiz, to be consider to its publication in BMC Complementary and Alternative Medicine.

In this version we have revised the manuscript and included line and page numbers.

We do hope you find our improved manuscript suitable for publication in this journal.

For further communication please contact to:

Dr. Juan Carlos Gálvez-Ruiz, Departamento de Ciencias Quimico-Biológicas, Universidad de Sonora, Blvd. Luis Encinas y Rosales S/N, 83000, Hermosillo, Sonora, Mexico. Phone and Fax:+52 662 259 21 63 and 64, e-mail: anisolborano@guayacan.uson.mx

Sincerely yours,

Dr. Juan Carlos Gálvez-Ruiz
Universidad de Sonora